Author's response to reviews

Title: Association of IL1A and IL1B loci with primary open angle glaucoma

Authors:

Suddhasil Mookherjee (suddhasilmookherjee@gmail.com)
Deblina Banerjee (banerjee.deblina@gmail.com)
Subhadip Chakraborty (subhadipiicb@gmail.com)
Antara Banerjee (banerjee.anta77@gmail.com)
Indranil Mukhopadhyay (indranilm100@gmail.com)
Abhijit Sen (contact@dristipradip.com)
Kunal Ray (kunalray@gmail.com)

Version: 2 Date: 17 July 2009

Author's response to reviews: see over
Dear Editor,

We are re-submitting the manuscript entitled “Association of IL1A and IL1B loci with primary open angle glaucoma” for consideration of its publication in BMC Medical Genetics as a Research Article, after checking the pdf of the submitted version, as we found minor errors in the Methods section and Table 3 of the earlier document.

The details are listed below:

1. **Page 4: Methods Section, Subsection: Selection of study subjects, 3rd paragraph, 2nd line**: “Age at diagnosis ranged from 42 to 88 years, with a mean ± standard deviation of 64 years ± 10 years” instead of “age at diagnosis ranged from 21 to 88 years, with a mean ± standard deviation of 62.11 years ± 11.88 years”.

2. **Page 15: Table 3, Description of the table**: “The p-values below 0.0167 are shown in bold letters and are considered to be significant” instead of “The p-values below 0.017 are shown in bold letters and are considered to be significant.”

We hope that you would find the manuscript worthy for reviewing it for evaluation of its publication in BMC Medical Genetics.

Sincerely,
Kunal Ray